Author's response to reviews

Title: Lung cancer presenting as a metastasis to the carpal bones: a case report.

Authors:

Giuseppe Rinonapoli Prof (grinonapoli@yahoo.it)
Auro Caraffa Prof (auro.caraffa@libero.it)
Renato Antenucci Prof (renato.antenucci@hotmail.it)

Version: 5 Date: 20 September 2012

Author's response to reviews:

As you requested, the Authors:
1) include the authors contribution section,
2) removed any text from figure seven which may reveal the patient’s identity,
3) corrected the grammatical structure of the below sentence:
   'The patient’s general condition was worsening because of respiratory problems, although we was submitted to chemotherapy'.